

Citation: Dermauw V, Van De Vijver E, Dorny P, Giorgi E, Ganaba R, Millogo A, et al. (2023) Geostatistical analysis of active human cysticercosis: Results of a large-scale study in 60 villages in Burkina Faso. PLoS Negl Trop Dis 17(7): e0011437. https://doi.org/10.1371/journal. pntd.0011437

Editor: Cristian A. Alvarez Rojas, Pontificia Universidad Catolica de Chile, CHILE

Received: September 27, 2022

Accepted: June 6, 2023

Published: July 26, 2023

Copyright: © 2023 Dermauw et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are included in the manuscript and its Supporting Information files.

Funding: The outcome data used for this work were generated in the context of the EFECAB project. This project was funded by the National Institute of Neurological Disorders and Stroke and the Fogarty International Center of the US National Institutes of Health under the Brain Disorders in the Developing World: Research Across the Lifespan **RESEARCH ARTICLE**

Geostatistical analysis of active human cysticercosis: Results of a large-scale study in 60 villages in Burkina Faso

Veronique Dermauw^{1*}, Ellen Van De Vijver², Pierre Dorny¹, Emanuele Giorgi³, Rasmané Ganaba⁴, Athanase Millogo^{5,6}, Zékiba Tarnagda⁷, Assana Kone Cissé⁷, Hélène Carabin^{8,9,10,11,12}

1 Department of Biomedical Sciences, Institute of Tropical Medicine, Antwerp, Belgium, 2 Department of Environment, Faculty of Bioscience Engineering, Ghent University, Ghent, Belgium, 3 Centre for Health Informatics, Computing, and Statistics, Lancaster University, Lancaster, United Kingdom, 4 AFRICSanté, Bobo-Dioulasso, Burkina Faso, 5 Centre Hospitalier Universitaire Sourô Sanou, Bobo-Dioulasso, Burkina Faso, 6 Université Ouagadougou, Joseph Ki-Zerbo, Ouagadougou, Burkina Faso, 7 Institut de Recherche en Sciences de la Santé, Bobo-Dioulasso, Burkina Faso, 8 Department of Pathology and Microbiology, University of Montreal, Montreal, Canada, 9 Centre de Recherche en Santé Publique (CReSP), Montreal, Canada, 10 Groupe de Recherche en Épidémiologie des Zoonoses et Santé Publique (GREZOSP), Montreal, Canada, 11 Department of Social and Preventive Medicine, University of Montreal, Montreal, Canada, 12 Department of Biostatistics and Epidemiology, University of Oklahoma Health Sciences Center, Oklahoma City, Oklahoma, United States of America

* vdermauw@itg.be

Abstract

Cysticercosis is a neglected tropical disease caused by the larval stage of the zoonotic tapeworm (*Taenia solium*). While there is a clear spatial component in the occurrence of the parasite, no geostatistical analysis of active human cysticercosis has been conducted yet, nor has such an analysis been conducted for Sub-Saharan Africa, albeit relevant for guiding prevention and control strategies. The goal of this study was to conduct a geostatistical analysis of active human cysticercosis, using data from the baseline cross-sectional component of a large-scale study in 60 villages in Burkina Faso. The outcome was the prevalence of active human cysticercosis (hCC), determined using the B158/B60 Ag-ELISA, while various environmental variables linked with the transmission and spread of the disease were explored as potential explanatory variables for the spatial distribution of T. solium. A generalized linear geostatistical model (GLGM) was run, and prediction maps were generated. Analyses were conducted using data generated at two levels: individual participant data and grouped village data. The best model was selected using a backward variable selection procedure and models were compared using likelihood ratio testing. The best individual-level GLGM included precipitation (increasing values were associated with an increased odds of positive test result), distance to the nearest river (decreased odds) and night land temperature (decreased odds) as predictors for active hCC, whereas the village-level GLGM only retained precipitation and distance to the nearest river. The range of spatial correlation was estimated at 45.0 [95%CI: 34.3; 57.8] meters and 28.2 [95%CI: 14.0; 56.2] km for the individual- and village-level datasets, respectively. Individual- and village-level GLGM unravelled large areas with active hCC predicted prevalence estimates of at least 4% in the southprogramme, grant R01NS064901 (to HC). The URL of the program is http://www.fic.nih.gov/ Programs/Pages/brain-disorders.aspx. HC is funded in part by the Canada Research Chair in Epidemiology and One Health (CIHR/IRSC CRC 950–231857). The URL of the program is http:// chairs-chaires.gc.ca. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

east, the extreme south, and north-west of the study area, while patches of prevalence estimates below 2% were seen in the north and west. More research designed to analyse the spatial characteristics of hCC is needed with sampling strategies ensuring appropriate characterisation of spatial variability, and incorporating the uncertainty linked to the measurement of outcome and environmental variables in the geostatistical analysis.

Trial registration: ClinicalTrials.gov; NCT0309339.

Author summary

Cysticercosis is a serious, yet neglected disease caused by the larval stage of a zoonotic tapeworm, prevalent in many developing countries, including Burkina Faso. Being able to predict where the disease occurs is essential for running targeted prevention and control activities. In our study, we aimed to describe whether human cysticercosis cases in three provinces in Burkina Faso were clustered, and investigated whether there was a link between this clustering and some land and weather variables. Finally, we aimed to generate high-resolution prediction maps for the occurrence of the infection. We found that there was clustering at 45 meters for the individual- and 28.2 km for the village-level datasets, respectively. Increasing rainfall and proximity to a river were linked with this clustering. The generated prediction maps indicated there were important cysticercosis hotspots in the study area, especially in the extreme south and north-west, where the disease is thought to be more important. Further research should expand the use of spatial techniques to predict the occurrence of cysticercosis, the results of which can aid in the design of intervention programmes.

Introduction

Cysticercosis is a neglected tropical zoonosis, acquired by ingestion of eggs shed by a human tapeworm (*Taenia solium*) carrier. Humans can become infected because of poor hand hygiene (faecal-oral route) or through the consumption of contaminated food or water [1,2]. Upon ingestion of the eggs, the larval forms of *T. solium* will migrate throughout the body and develop into cysticerci, fluid-filled cyst-like structures (human cysticercosis (hCC)). In humans, cysticercosis (NCC)) [3]. Neurocysticercosis has an important public health impact, as many cases suffer from its neurological manifestations, such as seizures, chronic headaches, increased intracranial pressure and even death [4].

A geostatistical analysis uses a specific set of methods to process georeferenced data that are inherently more similar when closer together. As such, these methods allow researchers to describe and model the spatial variability patterns of the data. Another important, if not the most important, objective of a geostatistical analysis is to predict values at unsampled locations through the creation of continuous surfaces from point data, also called interpolation, while also estimating the uncertainty linked to these predictions [5]. These techniques can be used for public health threats, to unravel the spatial structure of the variability in prevalence estimates, to identify environmental characteristics affecting the distribution of prevalence estimates, and to predict where a prevalence value exceeds a specific threshold [6]. The latter can be most useful in terms of the identifying target areas for intervention strategies.

Model-based geostatistical methods have been used for a number of neglected tropical diseases such as lymphatic filariasis [7], onchocerciasis [8,9], schistosomiasis [10–14] and soiltransmitted helminth infections [15–18]. Studies focussing on the spatial distribution of the infections caused by *T. solium* have so far unravelled relevant information about disease clustering [19–24] and distance-dependent relationships between conditions caused by the different life stages of the parasite [25–28]. Yet, up to now, only one study has conducted a geostatistical analysis for human infections caused by *T. solium*. Using data from a national baseline serosurvey in Colombia, Galipó et al. [29] determined that there was spatial correlation in seropositivity estimated up to approximately 140 km. However, that study used an ELISA applied to blood samples collected on filter paper to detect circulating *T. solium* cysticercus antibodies, its results thus point to exposure to the parasite. To our knowledge, no study has so far conducted a geostatistical analysis for active *T. solium* infections, nor has such an analysis been conducted for Sub-Saharan Africa. Thus, the aim of this work was to conduct a geostatistical analysis of active hCC, using data from 60 villages in three provinces of Burkina Faso.

Methods

Ethics statement

Ethical approval was obtained from the University of Oklahoma Health Sciences Center Institutional Review Board and the Centre MURAZ ethical review panel in Burkina Faso. The data used for this study were collected for a trial, registered with ClinicalTrials.gov, number NCT0309339. Written informed consent was obtained for all study participants. Parents consented in writing for children younger than 18 years and children older than 10 years were asked for their written assent.

Study design

The field data used in this study were collected in the context of a large cluster randomizedcontrol trial (cRCT), called EFECAB, investigating the effect of an educational intervention on cumulative incidence of active hCC [30–33]. More specifically, currently reported data were collected during the baseline cross-sectional component of this trial, conducted between February 2011 and January 2012 (STROBE checklist, <u>S1 STROBE</u> checklist).

The study was conducted in 60 villages in three provinces of Burkina Faso (Fig 1). The inclusion and exclusion criteria, selection procedures for study provinces, departments, villages, concessions (i.e., a group of households living in a compound), households, and participants, and rationale for sample sizes, have been described elsewhere [30–33]. Briefly, the study provinces, corresponding together to 4.9% of the country's total area, were selected due to their large pig population (Boulkiemdé and Sanguié) or neighbouring position and local reports of humans feeding stool to pigs (Nayala). All pig raising departments were included in the study, and in each department, two villages meeting the eligibility criteria for the cRCT namely having a population of at least 1000 people (2006 census), being present on the map of the *Institut Géographique du Burkina* (year 2000), and separated from the other selected villages by at least 5 km, were randomly selected. Capitals of the region or province were excluded as well as villages located within 20 km of the large cities Koudougou or Ouagadougou or on a national or provincial road.

Participants

In each village, concessions were selected using a stratified random sampling approach, with the different pig production types (reproductive sows, piglets or no pigs) as strata [30]. In each village, 80 concessions were randomly selected (at least 10 concessions raising sows, at least 30



Fig 1. Location of Burkina Faso in Africa, and of the study area in Burkina Faso. (Study provinces: N = Nayala, S = Sanguié, B = Boulkiemde) (https://www. diva-gis.org/gdata).

concessions raising piglets aged 12 months or younger). A handheld GPS device was used to record the geographical coordinates at the centre of each concession. After asking the chief of each selected concession for consent to participate in the study, one household per concession was randomly sampled and the household chief asked for consent to participate. Finally, one eligible person from the selected household was randomly selected and asked for consent to participate. Individuals were eligible if they were at least 5 years old, had resided in the village during the past year, and intended to remain there for the next three years.

Eligible individuals (one person per household) were invited to participate in the study, this entailed providing a blood sample for the serological component of the study and answering a screening questionnaire. Recruitment continued until 60 people consented to participate in each village. Additionally, 20 eligible individuals per village, who refused to participate to the serological compound, were invited to only participate in the screening questionnaire. The screening questionnaire investigated the presence of symptoms that could be linked to NCC (i.e. chronic worsening severe headaches, seizures and/or epilepsy) as well as socio-demographic information and knowledge about cysticercosis and epilepsy. Participants who screened positive to the presence of symptoms suggestive of NCC were invited to provide a blood sample to help with the diagnosis of NCC. Parents consented for children younger than 18 years and children older than 10 years were asked for their assent.

Outcome data

Each of the participants was sampled for blood by means of venipuncture of the antebrachium vein, using a syringe and 10 ml serum gel tubes. Once collected, the tubes were cooled and

transported to the processing unit. There, serum was collected, and stored at -20°C until analysis.

Using the B158/B60 Ag-ELISA, serum samples were tested for the presence of excretory/ secretory circulating antigens of *T. solium* cysticerci, i.e. active cysticercosis [34]. The optical density (OD) of each serum sample was compared with the mean OD of eight negative reference human sera samples at a probability level of p = 0.001 to determine the test result. In an earlier study conducted in Ecuador, this B158/B60 Ag-ELISA was reported to have a sensitivity of 90% (95% Bayesian Credible Interval (BCI): 80–99%) and a specificity of 98% (95%BCI: 90– 99%) to detect active infection [35]. The Ag-ELISA result (i.e. positive or negative), represented the occurrence of active hCC, and the prevalence of active hCC was considered the outcome variable of the study. All hCC survey data were entered in an Excel file.

Environmental data

A list of environmental variables considered potentially linked with the transmission and spread of hCC based on earlier work on *Taenia* spp. and other helminths [36] was drafted and explored as potential explanatory variables for the spatial distribution of *T. solium*. Datasets for the environmental variables were retrieved from online accessible data sources (Tables 1 and S1). In case datasets for multiple time points were available, it was opted to go for a file that contained data from around the mid-point of the data collection period (i.e. July 2011).

The environmental datasets were clipped to the boundaries of the three study provinces to ensure faster processing of files. The coordinate reference system of the country boundaries and all environmental datasets were converted to the Universal Transverse Mercator 30N zone to allow distance measurements to be expressed in meters. The file with existing waterlines was additionally used to create a dataset ($1 \text{ km} \times 1 \text{ km}$) containing distance at each point to the nearest river. Furthermore, a prediction grid with cells of $1 \text{ km} \times 1 \text{ km}$ was generated.

Finally, environmental data were extracted at the living locations of the participants to the hCC serosurvey and at the centroids of the prediction grid. For continuous variables, this entailed bilinear interpolation, based on the values of the four nearest cells, while for categorical variables, simple interpolation was chosen, i.e. the value for the cell where a point was

Variable	Source	Date or time period	Unit	Spatial resolution
Country/province boundary	DIVA-GIS	1992	-	-
Potential evapotranspiration	CGIAR-CSI	July 2011	mm/month	1km
Elevation	CGIAR SRTM	-	m	1km
Land cover	MODIS-Terra	2011	-	0.5km
Land surface temperatures	MODIS-Terra	04-July-2011	Celsius	1km
NDVI	MODIS-Terra	12-July-2011	NDVI	1km
Precipitation	WorldClim	1970–2000, July	mm/month	1km
Water lines	DIVA-GIS	1992	-	-
Soil pH (0–5 cm)	ISRIC	-	-	250m
Soil sand (0–5 cm)	ISRIC	-	g/100g (%)	250m
Soil clay (0–5 cm)	ISRIC	-	g/100g (%)	250m
Soil silt (0–5 cm)	ISRIC	-	g/100g (%)	250m

Table 1. Sources for the environmental data considered as potential explanatory variables for the spatial distribution of T. solium.

CGIAR-CSI: Consultative Group for International Agricultural Research—Consortium for Spatial Information (CGIAR-CSI); CGIAR SRTM: CGIAR Shuttle Radar Topography Mission; DIVA-GIS: Data-Interpolating Variational Analysis—Geographic Information System; ISRIC: International Soil Reference and Information Centre; MODIS: Moderate Resolution Imaging Spectroradiometer; NDVI: Normalized Difference Vegetation Index; WorldClim: World Climate

https://doi.org/10.1371/journal.pntd.0011437.t001

included. Values were expressed in their original units by multiplying with the scale factor mentioned at each data source, temperatures were additionally converted from Kelvin to Celsius.

Data analysis

For the data analysis, two databases were used: i) a dataset including information at the individual level, ii) a dataset including the individual data grouped at the village level. The villagelevel dataset included the number of sampled individuals and the number of the participants testing positive using the Ag-ELISA per village, the mean coordinates of the village survey points, and the village mean of survey point values for the environmental variables.

Descriptive statistics were calculated for the outcome and environmental variables. Maps of the study area, location of the survey points and village prevalence estimates were created, so were maps of the study area for the environmental data. Spearman's ρ_s correlation coefficients were calculated to measure the association between sets of two environmental variables to detect potential issues of multicollinearity.

A variogram $\hat{V}(u)$ was calculated to describe the spatial continuity of the outcome variable both at the village- and individual-level as follows:

For the individual-level data, the variogram was calculated using the outcome value for each participant. For the village-level data, the empirical logit [37] of the outcome variable was first calculated:

$$y_i^* = \log\left(\frac{y_i + 0.5}{n_i - y_i + 0.5}\right) \tag{1}$$

with y_i , the number of participants with positive test results in village *i*, and n_i , the total number of participants samples in village *i*. The addition of 0.5 was included to accommodate for $y_i = 0$ and $y_i = n_i$, i.e. to avoid plus and minus infinity. The variogram $\hat{V}(u)$ was then calculated for the empirical logit y_i^* , using the following formula (6):

$$\hat{V}(u) = \frac{1}{2|N(u)|} \sum_{(h,k)\in N(u)} (y_h^* - y_k^*)^2$$
(2)

with N(u) being the number of data-pairs at lag distance u apart from each other; u is the lag distance between two points, and y_h^* and y_k^* are the empirical logit for the outcome variable at location h and k, respectively. Data pairs were first classified in distance bins and the empirical variogram was averaged for each bin.

For both datasets, a variogram model was then fitted to get a first estimate for the variogram parameters, using weighted least squares, with the weights being the number of pairs per bin, and the general-purpose Nelder–Mead approach for numerical optimization [38].

Next, and for each dataset, the association between the outcome variable and each environmental variable was explored, using scatter plots and generalized linear models (GLM) (for detailed procedure see <u>S1 Text</u>). Briefly, the final selection of environmental variables was done using a backward stepwise approach, where models were compared using the likelihoodratio test. Once the final model had been selected, the presence of residual spatial correlation was investigated using a using a Monte Carlo method (see <u>S1 Text</u> for details).

The selected environmental variables were then included in a generalized linear geostatistical model (GLGM) formulated with the following structure (6):

$$log\left\{\frac{p(x_i)}{1-p(x_i)}\right\} = d(x_i)^t \beta + S(x_i) + Z_i$$
(3)

with $p(x_i)$ being the probability for a positive test result; $d(x_i)^t\beta$, the transposed column vector of the selected explanatory variables d_i at point x_i with regression coefficients β . The spatial term, $S(x_i) \sim N(0, \sigma^2)$ represents a dependent stationary isotropic Gaussian process, i.e. a Gaussian process with constant variance (σ^2 compares to the sill in classical geostatistical terminology) of which the covariance only depends on the distance between points. The correlation between $S(x_i)$ and $S(x_j)$, the spatial component at locations *i* and *j*, is defined by the correlation function $\rho(u)$. Several correlation functions have been suggested including the exponential, spherical and Matérn families. Here, we assume that $\rho(u; \phi) = \exp\left(-\frac{u}{\phi}\right)$, where *u* is the lag distance between points x_i and x_j , with $u \ge 0$; and ϕ a scale factor with $\phi > 0$. For this exponential correlation function, the practical range u_p (i.e. the lag distance where 95% of the sill is reached, or where $\rho(u)$ has decayed to 0.05) is $u_p \approx 3 \times \phi$. Z_i is a mutually independent Gaussian process, i.e. the spatial variation at a distance below the minimum observed distance, or random variation due to measurement error, with $Z_i \sim N(0, \tau^2)$, where τ^2 compares to the nugget in classical geostatistical terminology.

Three GLGMs were fitted, M_1 and M_2 , an individual- and a village- level model, respectively, with the covariates selected in the final individual and village level GLM (S3 Table) as well as an additional GLGM, M_3 , which had the same covariates as the village-level GLM, to allow comparison of the final prediction maps generated by the village- and individual-level data.

Parameter estimation was performed using the Monte Carlo maximum-likelihood (MCML) method which is detailed in [39] and [40]. The algorithm used to fit the GLGM (\mathcal{M}_1 , \mathcal{M}_2 , \mathcal{M}_3) had the GLM estimates for the coefficients (S3 Table), and the weighted least squares estimates for the variogram parameters σ^2 and ϕ , as starting values, while τ^2 was fixed at 0 (as estimated as such in the variogram analysis). Overfitting was assessed by investigating the correlation matrix of the regression coefficient estimates, with a Pearson's ρ correlation coefficient approaching 1 or -1 considered indicative for overfitting [41].

The predictive target, T^* , was defined as the prevalence surface over the study area, A (6):

$$T^* = \left\{ p(x) = \frac{\exp\{T(x)\}}{1 + \exp\{T(x)\}} : x \in A \right\}$$
(4)

with p(x) being the probability for a positive test result; and $T(x) = d(x)^t \beta + S(x)$. Maps of the predicted prevalence and associated standard errors were created.

Additionally, both for the individual-level and village-level datasets, maps were generated for the exceedance of a set prevalence threshold, *l*. Based on our own experience, we set the prevalence threshold was at 5%, as this prevalence was deemed worth identifying for prioritising intervention campaigns. The probability for the prevalence to exceed this threshold (also called the 5% exceedance probability) was calculated as follows (6):

$$\mathcal{R}_{l}(T^{*}) = \{r(x) : P[W(x) > l|y] = r(x), x \in A\}$$
(5)

with *l* being the prevalence threshold; W(x), the predicted prevalence, $W(x) = g^{-1}{T(x)}$ with $g^{-1}{\cdot}$, the inverse link function in the GLGM; and r(x) the point-wise exceedance probability. This probability was depicted on the prediction grid with grid cells of 1 km × 1 km spatial resolution. Finally, to assess the compatibility of the chosen spatial correlation structure, (i.e. correct specification of nugget τ^2 , and scale factor, ϕ), a Monte Carlo procedure was used (6). The procedure is detailed in S2 Text.

All statistical procedures were conducted in R version 4.0.5 [42]. The statistical significance level was set at 5%.



Fig 2. Study area with village-level data points and prevalence (derived from the individual-level dataset), and histogram of the prevalence (https://www. diva-gis.org/gdata).

Results

Descriptive statistics

Data were available for a total of 3598 individuals living in 60 villages in three provinces in Burkina Faso (Fig 2). The north-south dimension of the study area was 154 km, while its eastwest dimension was 199 km, and its longest dimension was 195 km. Out of the 3598 individuals, 30 lacked information for the main outcome variable (i.e. Ag-ELISA result), 12 had no village name assigned, and 4 data points were co-located. Of the remaining 3551 participants, 115 (3.24%) tested positive in the Ag-ELISA for active hCC. The number of participants ranged between 52 and 66 per village, with a median of 59. Out of the 60 villages, 48 had positive cases, in those villages, the number of positive cases ranged from 1 to 7, with a median of 2. Overall, village prevalence estimates ranged from 0 to 11.5%, with a median of 1.85%. Villages



Fig 3. Correlation plot for individual-level (A) and village-level (B) environmental data in the study area. Evap: potential evapotranspiration, tempday: land surface temperature, day; tempnight: land surface temperature, night; ndvi: normalized difference vegetation index; rain: precipitation; soilpH: soil pH (0–5 cm); soilsilt: soil silt (0–5 cm); soilsand: soil sand (0-5cm); soilclay: soil clay (0–5 cm); waterdist: distance to the nearest river.

with higher prevalence estimates seemed primarily located in the south-east of the study area (Fig 2).

The environmental data showed considerable variation over the study area (S1 Fig and S2 Table), except for land cover, where type 31 (i.e. barren land) was predominant. For this reason, land cover was excluded from further analyses. A strong negative correlation existed between elevation and soil clay (individual-level data: Spearman $\rho = -0.65$), and between soil clay and soil sand ($\rho = -0.80$) (Fig 3A), whereas a positive correlation was observed between elevation and soil sand ($\rho = 0.69$). Correlations were similar for the village-level dataset (Fig 3B).

Variogram

Fig 4A illustrates the variogram for the individual-level data. When decreasing the maximum lag distance to 400 meters (18 bins, number of data pairs per bin: 31–1208), an increase in variogram values was observed when moving from zero to around 60 meter lag distance, suggesting spatial correlation at that scale. For the village-level data (Fig 4B), when the maximum lag distance was restricted to 30 km, one low variogram value was observed at the shortest lag distance for which the variogram was calculated, thereafter the values fluctuated around 0.60. The number of data pairs per bin (with 12 bins) varied between 4 and 52.





https://doi.org/10.1371/journal.pntd.0011437.g004

Model	Model-level	Parameter	Estimate	95%CI	<i>p</i> -value
\mathcal{M}_1	Individual	Intercept	-9.54	[-10.0;-9.03]	<0.001
		Precipitation (mm/month)	0.11	[0.10;0.11]	<0.001
		Distance to the nearest river (km)	-0.088	[-0.090;-0.086]	<0.001
		Land temperature, night (°C)	-0.47	[-0.48;-0.46]	<0.001
		$\log(\sigma^2)$	-3.63	[-3.67;-3.58]	
		$\log(\phi)$	-4.21	[-4.47;-3.95]	
M ₂	Village	Intercept	-20.4	[-33.4;-7.37]	<0.002
		Precipitation (mm/month)	0.10	[0.024;0.18]	0.011
		Distance to the nearest river (km)	-0.086	[-0.14;-0.035]	0.001
		$\log(\sigma^2)$	-1.19	[-1.62;-0.75]	
		$\log(\phi)$	2.24	[1.54;2.93]	
M ₃	Individual	Intercept	-24.0	[-24.4;-23.6]	<0.001
		Precipitation (mm/month)	0.12	[0.12;0.13]	<0.001
		Distance to the nearest river (km)	-0.080	[-0.082;-0.078]	<0.001
		$\log(\sigma^2)$	-3.63	[-3.67;-3.58]	
		$\log(\phi)$	-4.21	[-4.47;-3.95]	

Table 2.	Parameter estimation	tes in generalized lin	near geostatistical	model for individual-	and village-level data
1	- within the continue		Louis Meconderent	moutiliter number and	and image to tot dutte

95%CI: 95% confidence interval

https://doi.org/10.1371/journal.pntd.0011437.t002

For the individual-level data, the sill, σ^2 , representing the variance of the spatial effect, was estimated at 0.029; the scale factor ϕ at 0.015 km with an estimated practical range, the distance over which data were spatially correlated, of 0.045 km. The nugget, τ^2 , representing sampling and/or measurement error, was estimated at 0. For the village-level data, the weighted least squares estimates for σ^2 , ϕ , and τ^2 were 0.68, 4.50 km (estimated practical range: 13.5 km) and 0, respectively.

Covariate selection

Based on the backward variable selection strategy, the final GLM included precipitation, distance to the nearest river and night land temperatures for the individual-level dataset while only precipitation and distance to the nearest river were retained for the village-level dataset (<u>S3 Text</u>).

Geostatistical model

The fitted GLGM for the individual-level data, \mathcal{M}_1 , predicted that for each unit increase in precipitation (mm/month), the odds for a positive test result increased by 12% (exp(0.11) = 1.12 [95% confidence interval (CI): 1.11;1.12], 1.12–1 = 0.12), if the other covariates were kept constant (Table 2). Likewise, for each unit increase in distance to the nearest river (km), a decrease in the odds of 8% (exp(-0.088) = 0.92 [95%CI: 0.91;0.92], 1–0.92 = 0.08) and for each unit increase in night land temperature (°C), a decrease in the odds of 37% (exp(-0.47) = 0.63 [95%CI: 0.62;0.63], 1–0.63 = 0.37) were predicted, if the other covariates were kept constant. In the individual-level GLGM, \mathcal{M}_1 , and \mathcal{M}_3 , the variance of the spatial effect, σ^2 was estimated at exp(-3.63) = 0.027 [95%CI: 0.025;0.028], whereas the shape factor, ϕ was estimated at exp (-4.21) = 0.015 [95%CI: 0.011;0.019]. The practical range, u, for hCC spatial correlation, after accounting for the covariates, was thus $u\approx3\times\phi = 0.045$ km or 45 [95%CI: 34.3;57.8] meters. In other words, individual level data were estimated to be spatially correlated up to 45 meters.





Similar effect sizes of the environmental variables were observed using the village-level dataset. Indeed, the fitted GLGM, M_2 , predicted that for each unit increase in precipitation (mm/month), the odds for a positive test result was increased by 11% (exp(0.10) = 1.11 [95% CI: 1.02;1.20], 1.11–1 = 0.11), if the distance to the nearest river was kept constant. Moreover, for each unit increase in distance to the nearest river (km), a decrease in the odds for a positive test result by 8% (exp(-0.086) = 0.92 [95%CI: 0.87;0.97], 1–0.92 = 0.08) was predicted, if precipitation was kept constant. The variance of the spatial effect, σ^2 was estimated at exp(-1.19) = 0.30 [95%CI: 0.20;0.47], whereas the shape factor, ϕ was estimated at exp(2.24) = 9.39 [95%CI: 4.66;18.7]. The practical range, *u*, for hCC spatial correlation was thus $u\approx 3\times \phi = 28.2$ [95%CI: 14.0;56.2] km, or in other words, the village-level data were estimated to be spatially correlated up to 28.2 km. No overfitting was observed for any of the GLGM (all $|\rho_P| < 0.35$).

Prediction

Maps were drawn for the predicted hCC prevalence, standard error and probability of exceeding the 5% threshold (Figs 5–7). The dataset for night land temperatures had six missing





values, resulting in a blank empty pixel at six locations on the prediction maps based on the individual-level GLGM \mathcal{M}_1 (Fig 5A). The prediction maps for active hCC prevalence generated from the individual- and village-level GLGM, \mathcal{M}_1 , \mathcal{M}_2 and \mathcal{M}_3 (Figs 5A, 6A, 7A, respectively), identified important areas with hCC prevalence estimates of at least 4% in the southeast, as well as, albeit to a lesser extent, in the extreme south, and north-west of the study area, while patches of prevalence estimates below 2% were mainly apparent in the north and west.

The standard errors for the prevalence estimates exhibited a similar pattern as the predicted prevalences for all GLGM (Figs 5B, 6B, 7B), with higher standard errors in areas with higher prevalence estimates. In the maps based on the individual-level GLGM, M_1 and M_3 , the standard errors were distinctly lower, as compared to those found for the village-level GLGM, M_2 .

The map of the probability for the active hCC prevalence to exceed 5% (Figs 5C, 6C, 7C) confirmed the conclusions drawn from the predicted prevalence maps: the exceedance probability was highest in the south-east and north-west of the study area (patches of >75% probability). However, the village-level GLGM (M_2) produced a markedly smoother map of the exceedance probability as compared to the individual-level GLGM (M_1 , and M_3) maps. In





the village-level GLGM (M_2) map, patches of >25% probability could also be observed in the south-east and north-west of the study area. Moreover, in large parts of the remaining areas in the north, west and far east of the study area, the exceedance probability was consistently lower than 10%.

Discussion

This study aimed to conduct a geostatistical analysis for active hCC in Burkina Faso, an endemic country in Sub-Saharan Africa. This analysis allows for the spatial prediction of the prevalence of active hCC at unsampled locations, which is informative to identify those areas that would benefit most from targeted intervention programmes. Indeed, the analysis resulted in a high resolution map ($1 \text{ km} \times 1 \text{ km}$) of the predicted distribution of active hCC in the study area, the provinces Boulkiemdé, Sanguié, and Nayala, with important clusters of hCC in the south-east, as well as in the extreme south, and north-west. Moreover, the analysis indicated that the practical range for spatial correlation of survey values was very short for the individual-level dataset (45 meters), while rather large for the village-level dataset (28.2 km). The

difference between these two distances could be due to the spatial aggregation of the villagelevel dataset (i.e. by design, all villages were at least 6 km apart from each other). Overall, these distances are a lot shorter than the 140 km for the municipality-level data observed by Galipó et al. [29]. However, a positive test result in the diagnostic test in that study (Ab-ELISA) indicates exposure to the parasite, whereas in our study (Ag-ELISA), it indicates active or current infection. Indeed, the Ab-ELISA can detect both viable, degenerating and calcified cysticerci, but also past exposure to the parasite that didn't result in infection, and a resolved infection [35]. The Ag-ELISA on the other hand will only detect viable and early-stage degenerating cysticerci, as only those will excrete/secrete antigens [35]. Furthermore, the actual test performance of the antibody-ELISA run on dried blood spots, as used in the Galipó study [29] is unclear considering that the paper describing the method does not clearly justify or explain study groups [43]. Moreover, the study of Galipó et al. [29] was a national survey, sampling 133 out of 1122 municipalities, in 23 out of 32 of the country's departments, whereas we sampled all departments in three provinces. Overall, it is therefore not clear to what extent the observed ranges can be compared.

The results of our fitted geostatistical models demonstrated that increasing precipitation, and decreasing distance to the nearest river were associated with an increased probability for a positive test result, and thus the presence of active hCC. In the individual-level model, decreasing night land temperatures were additionally found to be associated with increased probability for a positive test result. These results are in line with earlier studies indicating that *Taenia* spp. eggs survive better in a moist environment, while high temperatures are detrimental for their survival [36]. As expected, the variability in the hCC prediction maps closely resembled those in the maps of the selected covariates. Moreover, for the individual-level GLGM, the standard errors of the predicted prevalences were markedly lower than those generated by the village-level GLGM, partly explained by the lack of high resolution data for the latter as compared to the former. Finally, the exceedance probability map generated based on the village-level GLGM was considerably smoother than those for the individual-level GLGM, which could be explained by the combined effect of lower data density, and the smoothness in the environmental data and in the the spatial structure adopted for the village-level GLGM.

The analysis also highlighted several challenges related to the use of the survey data on hCC for prediction mapping. For the individual-level modelling, a large (ungrouped) dataset was used (n = 3551), the analysis of which was associated with a relatively high computational burden. For the village-level modelling on the other hand, the number of data points was suboptimal (n = 60). Webster and Oliver [44] had pointed out that at least 100 observations are necessary to ensure reliable calculation of the variogram, while Journal and Huijbregts [45] have indicated that at least 30 to 50 data-pairs are necessary for each distance bin. In the current study, the 60 data points resulted in a low number of data-pairs available for the calculation of the variogram model parameter estimation. Moreover, the sampling design dictated that villages needed be at least 5 km apart to be included, thus hampering the investigation of spatial continuity at the short distances. For both analyses, the variogram analysis estimated the nugget, representing sampling and/or measurement error (background noise) at 0. In practice, however, the absence of a nugget seems unrealistic, as sampling and/or measurement error might be present due to several factors.

Furthermore, a stratified multi-level random sampling approach was chosen for the purpose of a cluster-randomized trial to be conducted in the study area, but it is not necessarily ideal for conducting a geostatistical analysis. At the two extremes of sampling designs for geostatistical modelling, are completely random designs (where point samples are collected randomly across the study area), relatively efficient for estimation purposes; and completely regular designs (where point samples are, for instance, collected at regular distances across the study area) for efficient spatial prediction [46]. The latter however assumes that the variogram model parameters are known, which is most often not the case (these parameters then need to be estimated from a single dataset), and for most geostatistical problems, a compromise is needed between both goals [6,47]. This was also the case in the present work, where the variogram model parameters were estimated from the same datasets used for prediction. It has been shown that the completely regular design with close pairs added is one of the most effective designs in this case (e.g., Diggle and Lophaven [48]), which is recommended for future geostatistical analyses for hCC.

A number of limitations with regard to the interpretation and validity of study results were also present. There were uncertainties related to the measurement of the outcome variable. Firstly, the applied diagnostic tool is not a perfect test, its performance is characterised by a sensitivity of 90% and specificity of 98%, thus impacting the estimation of the true prevalence [35]. Furthermore, the selection of participants was not random, rather they were selected from the concessions depending on pig type (i.e. sows, or piglets, although the levels of the variable were not mutually exclusive). The presence of pigs could be linked to presence of T. solium carriers and hence to the presence of hCC cases. In most villages, this should not have been a huge issue since most concessions actually had pigs, so sampling bias due to this factor was assumed to be minimal. It is also important to note that the survey data were considered as point data throughout the analysis, i.e. it was assumed that the spatial support of the observations was negligible compared to the total size of the study area, and implicitly also that there was no uncertainty in the location of these point data. However, for the data at hand, some reservations exist about the extent to which this assumption holds. Participants did not reside at a single point location only, but moved over a certain surface in space (spatial support). This surface was unknown and, moreover, differed in size and shape between different participants. For both datasets, this had an impact on the calculation of variogram values at short distances. For actual point data, the minimum distance for which a variogram value can be calculated is limited to the smallest sampling distance. For point data that are actually surface data, such as the data at hand, there can be an additional restriction due to the minimum area (size) represented in the data points [49].

Several simplifications and assumptions were also made, which could have impacted the quality of the environmental data used in the analysis. First, either yearly or mid-year datasets were often the only ones available, which resulted in differences in temporal support across the environmental variables. These choices also resulted in the assumption that there was no considerable temporal variation, or that the observation in the middle of the sampling period was sufficiently representative of the entire sampling period. Additionally, there was also a mismatch between time of measurement (i.e. temporal support) of the explanatory variables and the outcome data which came from a cross-sectional study conducted between February 2011 and January 2012. For the village-level dataset, the mean value of the sampled survey points was used as village-level value for each environmental variable, resulting in loss of within-village variability and an assumption that the mean was a representative summary statistic. For the soil variables, it was assumed that the top layer (0-5 cm depth) was the most relevant layer for transmission of T. solium. However, it is not inconceivable that eggs of the parasite can survive at larger depths, and that in case of erosion due to heavy rainfall, especially in combination with a considerable slope, these deeper layers could also contribute to the transmission. Finally, the analysis combined environmental and outcome values for surveys points in three provinces in Burkina Faso, thus assuming the structure of the spatial processes is similar in the three provinces. This could be further investigated in separate analyses for the three provinces. Overall, however, this study remains the only study of its type, and will guide further studies in the field.

Supporting information

S1 STROBE Checklist. (DOCX)

S1 Text. Covariate selection procedure: methodology. (DOCX)

S2 Text. Validation procedure. (DOCX)

S3 Text. Covariate selection procedure: results. (DOCX)

DOCX)

S4 Text. Validation results. (DOCX)

S1 Table. Detailed data sources for the environmental data considered as potential explanatory variables for the spatial distribution of *T. solium*. (DOCX)

S2 Table. Summary measures for environmental data extracted for individual- and villagelevel datasets.

(DOCX)

S3 Table. Fitted generalized linear models for the individual-level and village-level datasets.

(DOCX)

S4 Table. Fitted generalized linear mixed models for the individual-level and village-level datasets.

(DOCX)

S1 Fig. Maps of the environmental variables at the study area. (DOCX)

S2 Fig. Relationship between village-level environmental variables and empirical logit outcome (Part A). (Green = linear regression, blue = general additive model with penalized smoother).

(DOCX)

S3 Fig. Relationship between village-level environmental variables and empirical logit outcome (Part B). (Green = linear regression, blue = general additive model with penalized smoother).

(DOCX)

S4 Fig. Relationship between individual-level environmental variables and outcome (Part A). (Green = linear regression, blue = general additive model with penalized smoother). (DOCX)

S5 Fig. Relationship between individual-level environmental variables and outcome (Part B). (Green = linear regression, blue = general additive model with penalized smoother). (DOCX)

S6 Fig. Testing for residual spatial correlation for the village-level data (distance in km). (DOCX)

S7 Fig. Test the fit of the final geostatistical model for the village-level data (distance in km). (DOCX)

S1 Data. Village data. (XLSX)

Acknowledgments

We would like to thank the EFECAB field team, for their hard work collecting the survey data, and the study participants for their willingness to join the study.

Author Contributions

Conceptualization: Veronique Dermauw, Rasmané Ganaba, Athanase Millogo, Zékiba Tarnagda, Hélène Carabin.

Data curation: Veronique Dermauw, Hélène Carabin.

Formal analysis: Veronique Dermauw.

Funding acquisition: Pierre Dorny, Rasmané Ganaba, Athanase Millogo, Zékiba Tarnagda, Hélène Carabin.

Investigation: Veronique Dermauw, Rasmané Ganaba, Athanase Millogo, Zékiba Tarnagda, Hélène Carabin.

Methodology: Veronique Dermauw, Ellen Van De Vijver, Emanuele Giorgi, Rasmané Ganaba, Athanase Millogo, Zékiba Tarnagda, Hélène Carabin.

Project administration: Rasmané Ganaba, Athanase Millogo, Hélène Carabin.

Resources: Pierre Dorny, Rasmané Ganaba, Athanase Millogo, Zékiba Tarnagda, Assana Kone Cissé.

Software: Veronique Dermauw, Emanuele Giorgi.

Supervision: Ellen Van De Vijver, Pierre Dorny, Hélène Carabin.

Validation: Veronique Dermauw.

Visualization: Veronique Dermauw.

Writing - original draft: Veronique Dermauw.

Writing – review & editing: Veronique Dermauw, Ellen Van De Vijver, Pierre Dorny, Emanuele Giorgi, Rasmané Ganaba, Athanase Millogo, Zékiba Tarnagda, Assana Kone Cissé, Hélène Carabin.

References

- Carabin H, Traoré A. Taenia solium taeniasis and cysticercosis control and elimination through community-based interventions. Curr Trop Med Reports. 2014; 1(4):181–93. https://doi.org/10.1007/s40475-014-0029-4 PMID: 25544938
- Del Brutto OH. Human cysticercosis (Taenia solium). Trop Parasitol. 2013; 3(2):100–3. https://doi.org/ 10.4103/2229-5070.122103 PMID: 24470991
- García HH, Gonzalez A, Evans C, Gilman R, Cysticercosis Working Group in Peru. Taenia solium cysticerosis. Lancet. 2003; 362(9383):547–56.

- Carabin H, Ndimubanzi PC, Budke CM, Nguyen H, Qian Y, Cowan LD, et al. Clinical manifestations associated with neurocysticercosis: A systematic review. PLoS Negl Trop Dis. 2011; 5(5):e1152. https://doi.org/10.1371/journal.pntd.0001152 PMID: 21629722
- Goovaerts P. Geostatistical tools for characterizing the spatial variability of microbiological and physicochemical soil properties. Biol Fertil Soils. 1998; 27(4):315–34.
- 6. Diggle PJ, Giorgi E. Model-based Geostatistics for Global Public Health. Methods and Applications. Boca Raton, Florida: CRC Press, Chapman & Hall; 2019.
- Deshpande A, Miller-Petrie MK, Johnson KB, Abdoli A, Abrigo MRM, Adekanmbi V, et al. The global distribution of lymphatic filariasis, 2000–18: a geospatial analysis. Lancet Glob Heal. 2020; 8(9):e1186– 94. https://doi.org/10.1016/S2214-109X(20)30286-2 PMID: 32827480
- O'Hanlon SJ, Slater HC, Cheke RA, Boatin BA, Coffeng LE, Pion SDS, et al. Model-based geostatistical mapping of the prevalence of Onchocerca volvulus in West Africa. PLoS Negl Trop Dis. 2016; 10(1):1– 36. https://doi.org/10.1371/journal.pntd.0004328 PMID: 26771545
- Zouré HGM, Noma M, Tekle AH, Amazigo U V., Diggle PJ, Giorgi E, et al. The geographic distribution of onchocerciasis in the 20 participating countries of the African Programme for Onchocerciasis Control: (2) pre-control endemicity levels and estimated number infected. Parasites and Vectors. 2014; 7(1):1– 15. https://doi.org/10.1186/1756-3305-7-326 PMID: 25053392
- Koroma JB, Peterson J, Gbakima AA, Nylander FE, Sahr F, Magalhães RJS, et al. Geographical distribution of intestinal schistosomiasis and soil-transmitted helminthiasis and preventive chemotherapy strategies in Sierra Leone. PLoS Negl Trop Dis. 2010; 4(11). https://doi.org/10.1371/journal.pntd. 0000891 PMID: 21124881
- Schur N, Hürlimann E, Garba A, Traoré MS, Ndir O, Ratard RC, et al. Geostatistical model-based estimates of Schistosomiasis prevalence among individuals aged ≤20 years in West Africa. PLoS Negl Trop Dis. 2011; 5(6).
- Kabore A, Biritwum NK, Downs PW, Soares Magalhaes RJ, Zhang Y, Ottesen EA. Predictive vs. empiric assessment of schistosomiasis: Implications for treatment projections in Ghana. PLoS Negl Trop Dis. 2013; 7(3). https://doi.org/10.1371/journal.pntd.0002051 PMID: 23505584
- Schur N, Vounatsou P, Utzinger J. Determining treatment needs at different spatial scales using geostatistical model-based risk estimates of schistosomiasis. PLoS Negl Trop Dis. 2012; 6(9). <u>https://doi.org/</u> 10.1371/journal.pntd.0001773 PMID: 23029570
- Chammartin F, Houngbedji CA, Hürlimann E, Yapi RB, Silué KD, Soro G, et al. Bayesian risk mapping and model-based estimation of Schistosoma haematobium–Schistosoma mansoni co-distribution in Côte d'Ivoire. PLoS Negl Trop Dis. 2014; 8(12).
- Assoum M, Ortu G, Basáñez MG, Lau C, Clements ACA, Halton K, et al. Spatiotemporal distribution and population at risk of soil-transmitted helminth infections following an eight-year school-based deworming programme in Burundi, 2007–2014. Parasites and Vectors. 2017; 10(1):1–12.
- Yapi RB, Chammartin F, Hürlimann E, Houngbedji CA, N'Dri PB, Silué KD, et al. Bayesian risk profiling of soil-transmitted helminth infections and estimates of preventive chemotherapy for school-aged children in Côte d'Ivoire. Parasites and Vectors. 2016; 9(1):1–9.
- Oluwole AS, Ekpo UF, Karagiannis-Voules DA, Abe EM, Olamiju FO, Isiyaku S, et al. Bayesian geostatistical model-based estimates of soil-transmitted helminth infection in Nigeria, including annual deworming requirements. PLoS Negl Trop Dis. 2015; 9(4):1–15. https://doi.org/10.1371/journal.pntd. 0003740 PMID: 25909633
- Chammartin F, Guimarães LH, Scholte RGC, Bavia ME, Utzinger J, Vounatsou P. Spatio-temporal distribution of soil-transmitted helminth infections in Brazil. Parasites and Vectors. 2014; 7(1):1–16. <u>https://</u> doi.org/10.1186/1756-3305-7-440 PMID: 25230810
- Raghava MV, Prabhakaran V, Jayaraman T, Muliyil J, Oommen A, Dorny P, et al. Detecting spatial clusters of Taenia solium infections in a rural block in South India. Trans R Soc Trop Med Hyg. 2010; 104:601–12. https://doi.org/10.1016/j.trstmh.2010.06.002 PMID: 20638091
- Ngowi HA, Kassuku AA, Carabin H, Mlangwa JE., Malongo RSM, Mbilinyi BP, et al. Spatial clustering of porcine cysticercosis in Mbulu, northern Tanzania. PLoS Negl Trop Dis. 2010; 4(4):e652.
- Ron-Garrido L, Coral-Almeida M, Gabriël S, Benitez-Ortiz W, Saegerman C, Dorny P, et al. Distribution and potential indicators of hospitalized cases of neurocysticercosis and epilepsy in Ecuador from 1996 to 2008. PLoS Negl Trop Dis. 2015; 9(11):e0004236. <u>https://doi.org/10.1371/journal.pntd.0004236</u> PMID: 26580210
- 22. Madinga J, Kanobana K, Lukanu P, Abatih E, Baloji S, Linsuke S, et al. Geospatial and age-related patterns of Taenia solium taeniasis in the rural health zone of Kimpese, Democratic Republic of Congo. Acta Trop. 2016;1–10. https://doi.org/10.1016/j.actatropica.2016.03.013 PMID: 26996821

- Coral-Almeida M, Henriquez-Trujillo AR, Asanza S, Erazo C, Paucar M, Calvopiña M. Assessing the burden and spatial distribution of Taenia solium human neurocysticercosis in ecuador (2013–2017). PLoS Negl Trop Dis. 2020; 14(6):e0008384. https://doi.org/10.1371/journal.pntd.0008384 PMID: 32511228
- Li H, Zang X, Hu X, Abe EM, Qian M, Xue J, et al. Spatio-temporal distribution characteristics of cysticercosis from 2000 to 2014 in Dali, Yunnan province, China. Geospat Health. 2020; 15(2). https://doi.org/10.4081/gh.2020.815 PMID: 33461276
- Lescano A, García H, Gilman R, Guezala M, Tsang V, Gavidia C, et al. Swine cysticercosis hotspots surrounding Taenia solium tapeworm carriers. Am J Trop Med Hyg. 2007; 76(2):376–83. PMID: 17297051
- Lescano A, Garcia H, Gilman R, Gavidia C, Tsang V, Rodriguez S, et al. Taenia solium cysticercosis hotspots surrounding tapeworm carriers: Clustering on human seroprevalence but not on seizures. PLoS Negl Trop Dis. 2009; 3(1):e371. https://doi.org/10.1371/journal.pntd.0000371 PMID: 19172178
- Lescano A, Pray I, Gonzalez A, Gilman R, Tsang V, Gamboa R, et al. Clustering of necropsy-confirmed porcine cysticercosis surrounding Taenia solium tapeworm carriers in Peru. Am J Trop Med Hyg. 2019; 100(2):314–22. https://doi.org/10.4269/ajtmh.18-0296 PMID: 30560769
- 28. Pray IW, Ayvar V, Gamboa R, Muro C, Moyano LM, Benavides V, et al. Spatial relationship between Taenia solium tapeworm carriers and necropsy cyst burden in pigs. 2017;1–15.
- Galipó E, Dixon MA, Fronterrè C, Cucunubá ZM, Basáñez MG, Stevens K, et al. Spatial distribution and risk factors for human cysticercosis in Colombia. Parasites and Vectors. 2021; 14(1):1–15.
- Carabin H, Millogo A, Cissé A, Gabriël S, Sahlu I, Dorny P, et al. Prevalence of and factors associated with human cysticercosis in 60 Villages in three provinces of Burkina Faso. PLoS Negl Trop Dis. 2015; 9 (11):1–20. https://doi.org/10.1371/journal.pntd.0004248 PMID: 26588468
- Carabin H, Millogo A, Ngowi HA, Bauer C, Dermauw V, Koné AC, et al. Effectiveness of a communitybased educational programme in reducing the cumulative incidence and prevalence of human Taenia solium cysticercosis in Burkina Faso in 2011–14 (EFECAB): a cluster-randomised controlled trial. Lancet Glob Heal. 2018; 6(4):e411–25. https://doi.org/10.1016/S2214-109X(18)30027-5 PMID: 29530423
- Ngowi H, Ozbolt I, Millogo A, Dermauw V, Somé T, Spicer P, et al. Development of a health education intervention strategy using an implementation research method to control taeniasis and cysticercosis in Burkina Faso. Infect Dis Poverty. 2017; 6(1):95.
- 33. Dermauw V, Carabin H, Ganaba R, Cissé A, Tarnagda Z, Gabriël S, et al. Factors associated with the 18-month cumulative incidence of seroconversion of active infection with taenia solium cysticercosis: A cohort study among residents of 60 villages in Burkina Faso. Am J Trop Med Hyg. 2018; 99(4):1018– 27. https://doi.org/10.4269/ajtmh.18-0294 PMID: 30182917
- Dorny P, Brandt J, Geerts S. Immunodiagnostic approaches for detecting Taenia solium. Trends Parasitol. 2004; 20(6):259–61. https://doi.org/10.1016/j.pt.2004.04.001 PMID: 15147673
- Praet N, Rodriguez-Hidalgo R, Speybroeck N, Ahounou S, Benitez-Ortiz W, Berkvens D, et al. Infection with versus exposure to Taenia solium: what do serological test results tell us? Am J Trop Med Hyg. 2010; 83(2):413–5. https://doi.org/10.4269/ajtmh.2010.10-0121 PMID: 20682891
- **36.** Jansen F, Dorny P, Gabriël S, Dermauw V, Johansen MV, Trevisan C. The survival and dispersal of Taenia eggs in the environment: what are the implications for transmission? A systematic review. Parasites and Vectors. 2021; 14(1):1–16.
- Agresti A. Categorical Data Analysis. 3rd Editio. Vol. 47, Wiley Series in Probability and Statistics. New Jersey, USA: John Wiley & Sons, Inc.; 2013. 755–758 p.
- 38. Nelder J, Mead R. A simplex method for function minimization. Comput J. 1965; 7(4):308–13.
- **39.** Christensen O. Monte Carlo Maximum Likelihood in Model-Based Geostatistics. J Comput Graph Stat. 2004;13(3).
- 40. Giorgi E, Diggle PJ. PrevMap: An R package for prevalence mapping. J Stat Softw. 2017; 78(8).
- 41. Giorgi E, Fronterrè C, Macharia PM, Alegana VA, Snow RW, Diggle PJ. Model building and assessment of the impact of covariates for disease prevalence mapping in low-resource settings: to explain and to predict. J R Soc Interface. 2021; 18:20210104. https://doi.org/10.1098/rsif.2021.0104 PMID: 34062104
- R Core Team. R: A language and environment for statistical computing. Vienna: R Foundation for Statistical Computing; 2023.
- Corredor A, López MC, Duque S, Nicholls RS. Estandarización y evaluación de ELISA en eluidos de sangre seca recolectada en papel de filtro para el diagnóstico de cisticercocis. Biomédica. 1996; 16 (2):131.
- Webster R, Oliver MA. Geostatistics for Environmental Scientists. Second edi. Chichester: John Wiley & Sons; 2007. 1–315 p.

- 45. Journel AG, Huijbregts CJ. Mining Geo-statistics. London: Academic Press; 1978. 600 p.
- 46. Shmueli G. To explain or to predict? Stat Sci. 2010; 25(3):289–310.
- 47. Webster R, Lark M. Field sampling for environmental science and management. Oxon: Routledge; 2012.
- 48. Diggle P, Lophaven S. Bayesian geostatistical design. Scand J Stat. 2006; 33(1):53-64.
- **49.** Waller L, Gotway C. Applied Spatial Statistics for Public Health Data. Hoboke, NJ: John Wiley & Sons; 2004.